Check for updates

GOPEN ACCESS

Citation: Takeuchi R, Njenga SM, Ichinose Y, Kaneko S, Estrada CA, Kobayashi J (2019) Is there a gap between health education content and practice toward schistosomiasis prevention among schoolchildren along the shores of Lake Victoria in Kenya? PLoS Negl Trop Dis 13(8): e0007572. https://doi.org/10.1371/journal.pntd.0007572

Editor: Abdallah M. Samy, Faculty of Science, Ain Shams University (ASU), EGYPT

Received: December 5, 2018

Accepted: June 22, 2019

Published: August 19, 2019

Copyright: © 2019 Takeuchi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript.

Funding: This work was supported by STI-COPP Africa Program: Science, Technology and Innovation (STI) Cooperation Promotion Program in Africa (Program for Integrated Promotion of Social System Reform and Research and Development) from Japan Agency for Medical Research and development, AMED. The funders had no role in study design, data collection and RESEARCH ARTICLE

Is there a gap between health education content and practice toward schistosomiasis prevention among schoolchildren along the shores of Lake Victoria in Kenya?

Rie Takeuchi ^{1*}, Sammy M. Njenga², Yoshio Ichinose¹, Satoshi Kaneko^{1,3}, Crystal A. Estrada⁴, Jun Kobayashi⁴

1 Kenya Research Station, Institute of Tropical Medicine, Nagasaki University, Nairobi, Kenya, 2 Eastern and Southern Africa Centre of International Parasite Control, Kenya Medical Research Institute, Nairobi, Kenya, 3 Department of Eco-epidemiology, Institute of Tropical Medicine, Nagasaki University, Nagasaki, Japan, 4 Department of Global Health, Graduate School of Health Sciences, University of the Ryukyus, Okinawa, Japan

* zhuneilihui@gmail.com

Abstract

Despite provision of preventive measures against schistosomiasis such as mass drug administration (MDA), the prevalence of Schistosoma mansoni remains high in communities living near Lake Victoria. This study aimed to analyse the status of schistosomiasis, including its prevalence, health education, knowledge, attitudes, and practices (KAP) among pupils, and water use in schools in Mbita situated along the shores of Lake Victoria. Four primary schools were selected as target schools and pupils in classes six and seven were recruited as study participants. The prevalence of S. mansoni was examined by Kato-Katz method. Simultaneously, a KAP survey toward schistosomiasis was conducted among the pupils. Health education contents were extracted from textbooks. All primary schools in the study site were surveyed regarding how each secured water used for daily school life. The prevalence of S. mansoni was 56% and 36% in 2015 and 2016, respectively. 60-70% of pupils chose a correct answer for the mode of transmission. More than 70% of pupils answered that bathing in Lake Victoria causes Schistosoma infection; however, more than 70% of pupils bathed in Lake Victoria sometimes or every day. According to the science textbook, "avoiding contact with contaminated water" is the way to prevent schistosomiasis; however, 66% of schools asked pupils to bring water from Lake Victoria. The prevalence of S. mansoni among pupils remains high. Schoolchildren are taught to avoid contact with contaminated water but are often asked to fetch water from the lake. From the school health viewpoint, health education that reflects the social and cultural context of the community in the contents and teaching methods are needed. In addition to this, provision of sanitation infrastructure is needed. A comprehensive and innovative approach which harmonises central and local governments and other stakeholders, as well as community is important to prevent schistosomiasis.

analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

According to WHO, it is estimated that at least 206 million people required preventive treatment for schistosomiasis in 2016 and of those at least 90% live in Africa. Moreover, school-age children are most affected by this worm. In Kenya, despite preventive measures against schistosomiasis such as mass drug administration (MDA) in primary schools, it remains a major public health problem, especially along the shores of Lake Victoria. Our study revealed that the prevalence of Schistosoma mansoni was high among pupils despite their knowledge about schistosomiasis transmission. This can be attributed to lack of safe water supply and latrines. A comprehensive approach which harmonises central and local governments and other stakeholders, as well as community is needed to prevent schistosomiasis. For example, the combination of health education (in which schools play as focal point) with provision of water infrastructure, encouragement to use and construct toilets in both schools and community/households, snail control, and MDA with monitoring can be employed to control and prevent schistosomiasis. From the school health viewpoint, it is necessary to develop health education contents and teaching methods that reflect the social and cultural context of the community in order to improve their behaviour and change the social norm.

Introduction

Schistosomiasis is a parasitic disease caused by Schistosoma species. The parasite infects humans through skin penetration during contact with water that is contaminated with schistosome cercariae. Schistosomiasis is listed as a neglected tropical disease (NTD) by the World Health Organization (WHO); it is endemic in subtropical and tropical areas with most of the burden in sub-Saharan Africa [1]. In 2012, WHO reported that an estimated 3,179,000 years lived with disability (YLDs) were attributable to schistosomiasis [2]. High prevalence and intensity are observed among school-age children, adolescents, and young adults [3,4]. Since 1950, the World Health Assembly (WHA) has repeatedly discussed control and elimination of schistosomiasis and made several resolutions [5-9], and various measures, such as snail control, mass treatment, sanitary methods, and health education, have been conducted by several organizations worldwide [10]. In 2002, WHO recommended three components for schistosomiasis control in school: drug treatment, sanitary improvement, and health education. All three components interrupt the life cycle of schistosomes [11]. To determine the success of a control programme, a knowledge, attitude, and practice (KAP) survey which can collect information on what is known, believed, and done toward the topic of interest among the target population is useful [12]. Moreover, it is essential to know the social, cultural, and behavioural determinants which can fill the gap between knowledge and practice [13]. So far, several studies on KAP toward schistosomiasis have been conducted in various African countries among various participants. Most of them found that both adult and child participants did not have enough knowledge and good practices, and this reality may disrupt schistosomiasis control [14 - 18].

In Kenya, urinary and intestinal schistosomiasis due to *S. haematobium and S. mansoni* infection are observed and highly endemic in Coast region and Western Kenya, respectively [19]. The Government of Kenya launched the National School-Based Deworming Programme (NSBDP) in 2009 through a collaboration between the Ministry of Education and Ministry of Health [20]. In terms of sanitary improvement, water, sanitation and hygiene (WASH) activities in schools and/or communities have been conducted in several parts of Kenya by several

organisations [21–24]. Despite implementation of preventive interventions such as mass drug administration (MDA) and WASH, the prevalence of *S. mansoni* in Mbita Sub-county along the shores of Lake Victoria has remained high [25–27]. This study aimed to analyse the status quo of schistosomiasis from a school health viewpoint. Specifically, this study described the prevalence, knowledge, attitude, and practice toward schistosomiasis among school children, health message through education, water use in primary schools enrolled in a school health project funded by Japan International Cooperation Agency (JICA), and the gap between messages from current preventive measures and actual practices.

Methods

Ethics statement

This study was approved by the Scientific and Ethics Review Unit, Kenya Medical Research Institute (SSC Protocol No. 2916). Permission for research activities in schools was given by the local education office. Written assent and written consent from pupils and their guardians, respectively, were obtained before commencement of data collection.

Study site

This study was conducted in four locations along the shores of Lake Victoria, in Mbita Subcounty, Homa Bay County, Kenya. The population of the study site in 2015 was 62,954, with fishery in Lake Victoria being the main source of income. In 2016, 3.3% of households had tap water and according to the Health and Demographic Surveillance System (HDSS) initiated by the NUITM-KEMRI project (Institute of Tropical Medicine, Nagasaki University, Japan and Kenya Medical Research Institute, Kenya), there were nine functional boreholes available in the area. The distribution of households having tap water, functional borehole, and target schools are shown in Fig 1. According to a school health project which has been conducted in the same area by the Institute of Tropical Medicine, Nagasaki University and funded by JICA since September 2012, there were 93 primary schools, including both public and private, complete and incomplete in the study area in 2016.

Study participants

The sample size was calculated using Epi Info 7 (CDC, USA). Using the sample size calculation method for a population survey with four clusters, the calculated sample size was 67 for each cluster, giving a total sample size of 268.

One school from each location in the study area, giving a total of four schools, were randomly selected through the following procedure: first, schools which already received or were currently receiving any intervention from other organizations/groups were excluded. The remaining schools were then assigned individual numbers according to locations, with the numbers for each location selected randomly through draw lots. To examine the prevalence of *S. mansoni* and *S. haematobium* and KAP toward schistosomiasis, pupils in class six and class seven were recruited to participate in this study. All 93 schools in the study area were invited to provide information on how they obtained water being used in schools.

Data collection

The presence of *S. mansoni* and *S. haematobium* infection was examined by Kato-Katz and urine filtration methods, respectively, by skilled laboratory technicians [28,29]. To increase sensitivity of the Kato-Katz method, stool samples collected over three consecutive days were examined for *S. mansoni* eggs. The presence of soil-transmitted helminths (STHs) was also



Fig 1. Study site and target schools. A green rectangle in the national map indicates the position of the area shown in the main map frame. Red stars in the main frame indicate target schools. Blue rhombuses indicate boreholes. Brown small dots indicate households that do not have piped water. Cream small dots indicate households that have piped water. Base map data was downloaded from USGS LandsatLook https://landsatlook.usgs.gov/viewer.html Kenya map image was downloaded from CIA World Factbook https://www.cia.gov/library/publications/the-worldfactbook/graphics/maps/large/ke-map.gif.

https://doi.org/10.1371/journal.pntd.0007572.g001

recorded. Urine samples were collected once. Stool samples were collected in the morning, from 7:30 to 8:00 a.m. before class would start, and urine samples were collected during school morning break which was scheduled from 11:00 a.m. to 11:30 a.m. Two slides were prepared from each sample and examined by two independent microscopists to detect *S. mansoni* or *S. haematobium* eggs. Intensity of infection was calculated as the number of eggs per gram of feces (epg) or per 10 ml of urine and categorised as light (1–99 epg), moderate (100–399 epg), or heavy (\geq 400 epg) according to WHO guideline [30]. Treatment with praziquantel were given to all schistosomiasis-positive pupils who participated in the study.

A schistosomiasis KAP survey was conducted among the pupils using a paper-based, closeended questionnaire. The questions that were asked included cause of infection, symptoms, prevention, and risk behaviours. Contents related to health education against schistosomiasis was extracted from primary school science textbooks by the researchers. Additionally, secondary data obtained from teacher interviews for a previous school health project were also used in the present study. Information that were obtained from the teacher interviews include how the water for daily school use such as drinking, washing hands, cleaning facilities, and school gardening was acquired. The stool and urine examinations and KAP survey among pupils were conducted twice—in October 2015 and September 2016, while the contents extraction of health education textbooks was carried out between October 2015 to September 2016. Finally, the survey on water use in schools was done in October 2016.

Table 1. Number of study participants.

		2015			2016		
	N	Boys: n (%)	Girls: n (%)	N	Boys: n (%)	Girls: n (%)	
School 1	73	41 (56)	32 (44)	70	34 (49)	36 (51)	
School 2	43	29 (67)	14 (33)	53	27 (51)	26 (49)	
School 3	77	48 (62)	29 (38)	72	39 (54)	33 (46)	
School 4	81	47 (58)	34 (42)	79	45 (57)	34 (43)	
Total	274	165 (60)	109 (40)	274	145 (53)	129 (47)	

https://doi.org/10.1371/journal.pntd.0007572.t001

Data analysis

Results of the laboratory exam and KAP survey in 2015 and 2016 were described using descriptive statistics while logistic regression modelling was used to explore the association between attitude and practice and to identify the potential risk factors of *Schistosoma* infection. 2015 and 2016 data were integrated for analysis of association and risk factors. All statistical analyses were performed using Stata 13.0 computer package (StataCorp., USA).

Results

Enrolment and characteristics of participants

<u>Table 1</u> shows the demographic characteristics of the study participants. 274 pupils (165 boys and 109 girls) participated in 2015. Similarly, 274 (145 boys and 129 girls) pupils also participated in the study in 2016 (<u>Table 1</u>). Mean age in year were 13.6 ± 1.2 in 2015 and 13.4 ± 1.1 in 2016.

Prevalence and intensity of S. mansoni, S. haematobium and STHs

A total of 218 (133 boys and 85 girls) and 219 (118 boys and 101 girls) pupils were successfully examined for *S. mansoni* and STH on three consecutive days in 2015 and 2016, respectively. The prevalence of *S. mansoni* infection was 56.4% in 2015 and 35.6% in 2016. The percentage of pupils with light, moderate, and heavy infection were 32.6%, 16.5%, and 7.3% in 2015 and 23.3%, 9.6%, and 2.7% in 2016, respectively (Table 2). The prevalence of STH infections was low both in 2015 and 2016; specifically, the prevalence of hookworm, *Ascaris lumbricoides*, and *Trichuris trichiura* were 2.8%, 0.5%, and 0.9% in 2015. Only *A. lumbricoides* was detected in 2016 at 0.5% prevalence. No *S. haematobium* infection was detected.

KAP on schistosomiasis

Table 3 summarizes the KAP results. More than 75% of pupils in both 2015 and 2016 answered that blood in urine and blood in stool were main signs of schistosomiasis. However, more than

	2015				2016					
	N	Positive: n (%)	Light: n (%)	Moderate: n (%)	Heavy: n (%)	N	Positive: n (%)	Light: n (%)	Moderate: n (%)	Heavy: n (%)
Overall	218	123 (56.4)	71 (32.6)	36 (16.5)	16 (7.3)	219	78 (35.6)	51 (23.3)	21 (9.6)	6 (2.7)
By school										
School 1	59	38 (64.4)	25 (42.4)	7 (11.9)	6 (10.2)	62	29 (46.8)	21 (33.9)	7 (11.3)	1 (1.6)
School 2	39	37 (94.9)	19 (48.7)	12 (30.8)	6 (15.4)	48	27 (56.3)	14 (29.2)	8 (16.7)	5 (10.4)
School 3	52	2 (3.9)	2 (3.9)	0 (0)	0 (0)	51	0 (0)	0 (0)	0 (0)	0 (0)
School 4	68	46 (67.7)	25 (36.8)	17 (25.0)	4 (5.9)	58	22 (37.9)	16 (27.6)	6 (10.3)	0 (0)

Table 2. Rate and intensity of Schistosoma mansoni infection among pupils.

https://doi.org/10.1371/journal.pntd.0007572.t002

	2015 n (%)	2016 n (%)
Knowledge		
Signs and symptoms (multiple answer)		
Abdominal pain	93 (35.2)	91 (34.6)
Diarrhoea	86 (32.6)	105 (40.0)
Blood in urine	234 (88.6)	232 (88.2)
Blood in stool	202 (76.5)	226 (85.9)
Severe fever	77 (29.2)	89 (33.8)
Itching	88 (33.3)	85 (32.3)
<u>Transmission</u> [*] ($N_{15} = 185, N_{16} = 229$)		
Eating raw/undercooked fish/meat	7 (3.8)	7 (3.1)
Eating unwashed fresh vegetables	10 (5.4)	17 (7.4)
Contact with contaminated water	110 (59.5)	159 (69.4)
Walking barefoot on contaminated soil	58 (31.3)	46 (20.1)
Prevention (multiple answer)		
Cooking fish/meat well	50 (18.9)	33 (12.6)
Washing fresh vegetables	69 (26.1)	57 (21.7)
Avoiding contact with contaminated water	149 (56.4)	221 (84.0)
Using toilet for urination/defecation	146 (55.3)	129 (49.1)
Wearing sandals	84 (31.8)	99 (37.6)
Using protective boots and gloves when we contact with water from lake	196 (74.2)	181 (68.8)
Washing hands before eating	101 (38.3)	88 (33.5)
Washing hands after using toilets	108 (40.9)	117 (44.5)
Source of information (multiple answer)		
Hospital/health centre	202 (76.5)	211 (80.2)
School teachers	247 (93.6)	242 (92.0)
Family	80 (30.3)	99 (37.6)
Friends/neighbours	77 (29.2)	96 (36.5)
CHV	27 (10.2)	46 (17.5)
Main sign of bilharzia in Mbita [*] $(N_{16} = 245)$		
I do not know.	No data	9 (3.7)
Blood in urine		139 (56.7)
Blood in stool		97 (39.6)
Attitude		
Having an opportunity to be infected with bilharzia		
Strongly disagree	97 (36.7)	101 (38.4)
Disagree	40 (15.1)	32 (12.2)
Neutral	15 (5.7)	15 (5.7)

Table 3. Results of knowledge, attitude and	practice towards schistosomiasis	by year (N	$N_{15} = 264, N_{16} = 263$
---	----------------------------------	------------	------------------------------

(Continued)

Table 3. (Continued)

	2015 n (%)	2016 n (%)
Agree	44 (16.7)	21 (8.0)
Strongly agree	68 (25.8)	94 (35.7)
Having an opportunity to be infected with bilharzia by bathing/fetching water/washing	No data	
clothes or utensils at Lake Victoria		
Strongly disagree		42 (16.0)
Disagree		15 (5.7)
Neutral		13 (4.9)
Agree		19 (7.2)
Strongly agree		174 (66.2)
Practice		
Bathing at Lake Victoria		
Everyday	148 (56.1)	132 (50.2)
Sometimes	61 (23.1)	57 (21.7)
Rarely	16 (6.0)	29 (11.0)
Never	39 (14.8)	45 (17.1)
Fetching water from Lake Victoria		
Everyday	139 (52.7)	150 (57.0)
Sometimes	84 (31.8)	74 (28.1)
Rarely	22 (8.3)	22 (8.4)
Never	19 (7.2)	17 (6.5)
Washing clothes/utensils at Lake Victoria		
Everyday	86 (32.6)	88 (33.5)
Sometimes	104 (39.4)	96 (36.5)
Rarely	22 (8.3)	30 (11.4)
Never	52 (19.7)	49 (18.6)
Using toilet at school		
Not at all	10 (3.8)	5 (1.9)
Sometimes	31 (11.7)	25 (9.5)
Every time	223 (84.5)	233 (88.6)
Using toilet at home		
Do not have a toilet at home	20 (7.6)	15 (5.7)
Not at all	20 (7.6)	19 (7.2)
Sometimes	48 (18.2)	52 (19.8)
Every time	176 (66.6)	177 (67.3)
Playing (swimming) at Lake Victoria	No data	
Everyday		58 (22.0)
Sometimes		93 (35.4)
Rarely		30 (11.4)
Never		82 (31.2)

*Participants who selected multiple answers were excluded.

https://doi.org/10.1371/journal.pntd.0007572.t003

	Number of S. mansoni positive (%)	Crude Odds Ratio (95% CI)
Bathing at lake		
Never	22 (34.4)	1
Rarely	9 (27.3)	0.72 (0.28-1.80)
Sometimes	30 (31.6)	0.88 (0.45-1.73)
Everyday	135 (58.2)	2.66 (1.49-4.74)
Fetching water from lake		
Never	10 (33.3)	1
Rarely	15 (45.5)	1.67 (0.60-4.63)
Sometimes	47 (35.9)	1.12 (0.48-2.59)
Everyday	124 (53.9)	2.34 (1.05-5.22)
Washing clothes/utensils at lake		
Never	25 (32.1)	1
Rarely	14 (33.3)	1.06 (0.48-2.36)
Sometimes	73 (44.0)	1.68 (0.95-2.93)
Everyday	84 (60.9)	3.30 (1.84-5.92)
Playing (swimming) at lake*		
Never	20 (31.3)	1
Rarely	5 (23.8)	0.69 (0.22-2.14)
Sometimes	25 (33.8)	1.12 (0.55-2.30)
Everyday	26 (49.1)	2.12 (1.00-4.51)

Table 4. Association between practice and *Schistosoma* infection (N = 424, *N = 212).

2015 and 2016 data were integrated for risk factors analysis.

*Data is available only for 2016.

https://doi.org/10.1371/journal.pntd.0007572.t004

half of the pupils wrongly chose "blood in urine" as a main sign of dominant schistosomiasis in Mbita (data is available only for 2016). Nearly 60–70% of pupils in 2015 and 2016 correctly answered "contact with contaminated water" as a transmission mode. In terms of prevention, "Using protective boots and gloves when we contact with water from lake", "avoiding contact with contaminated water" and "using toilet for urination/defecation" were the top three major answers for both years. Pupils cited schoolteachers (more than 90%) or hospital/health centre (about 80%) as their main sources of information. In terms of attitude, about half of pupils chose "strongly disagree" or "disagree" to having the possibility to get infected with schistosomiasis in daily life in both years. On the other hand, more than 70% of pupils "strongly agreed" or "agreed" to possibly getting infected with schistosomiasis by bathing/fetching water/washing clothes or utensils at Lake Victoria. Attitude toward infection opportunity during daily life and activities at the lake were statistically different (p < 0.001 in 2016 by χ^2 test). In terms of practice, more than 70%, 80%, and 70% of pupils bathed, fetched water and washed clothes/utensils, respectively at Lake Victoria every day or sometimes in both years. When asked about sanitation habits, about 85% of the pupils reported that they regularly use toilet while in school, but the figure statistically decreased to about 67% while at home in both years (p = 0.004 in 2015, p < 0.001 in 2016 by χ^2 test).

Risk factors for schistosomiasis were bathing in the lake (OR for everyday = 2.66, 95%CI: 1.49–4.74), fetching water from the lake (OR for everyday = 2.34, 95%CI: 1.05–5.22), washing clothes/utensils in the lake (OR for everyday = 3.30, 95%CI: 1.84–5.92) and playing/swimming in the lake (OR for everyday = 2.12, 95%CI: 1.00–4.51) (Table 4).

A positive association between bathing practice and attitude toward infection opportunity in daily life was also observed in the study (OR for everyday = 2.27, 95%CI: 1.37-3.77; OR for sometimes = 1.92, 95%CI: 1.09-3.40) (Table 5).

PLOS	NEGLECTED TROPICAL DISEASES
-------------	--------------------------------

	Agree to having opportunity of infection with schistosoma n (%)	Odds Ratio	Overall <i>p</i> -value
Bathing at lake			p = 0.0047
Never	26 (31.0)	1	
Rarely	15 (33.3)	1.18 (0.56-2.50)	
Sometimes	53 (44.9)	1.92 (1.09-3.40)	
Everyday	133 (47.5)	2.27 (1.37-3.77)	

Table 5. Association between practice of bathing and attitude toward opportunity of infection with *Schistosoma* during daily life using ordered logistic regression modelling (N = 527).

Attitude was regrouped into the following three categories: agree (strongly agree + agree), neutral, and disagree (disagree + strongly agree).

https://doi.org/10.1371/journal.pntd.0007572.t005

Education on schistosomiasis (contents of textbook)

In Kenya, an independent health subject was not included in the curriculum; thus, schistosomiasis was taught in the science subject. Pupils in class six learn about schistosomiasis in the science subject as "bilharzia"–a water-borne disease. Two kinds of textbooks published by different publishers were available in the study area, and their use depended on preference of the school. Both textbooks covered cause, transmission, and prevention. However, one of the textbooks wrongly described that drinking contaminated water is a mode of transmission, and the other missed the symptoms of intestinal schistosomiasis (<u>S1 Table</u>).

Water use in primary schools

Of the 93 schools, 71 (76%) depended on Lake Victoria as a water source for daily school life (Fig 2). 54% of the schools had at least one rainwater harvesting tank and used rainwater during rainy season. 15 (16%) schools had tap water, but of those 11 schools used another water source in combination due to unstable tap water supply. 61 of 93 (66%) schools asked pupils to fetch/bring water for use in schools and of those, 53 schools had Lake Victoria as the nearest water source (Fig 3). The other eight schools either used water from a dam, stream, or a bore hole near the school. The main purposes of water use were for hand washing and drinking. Several schools also used water for school gardening purposes.

Discussion

The prevalence of *S. mansoni* infection among pupils apparently changed in two years. It seemed to decrease; however, it has not decreased significantly when compared to the prevalence reported by previous studies in Mbita district [25–27]. Re-infection rate reported in previous studies in Kenya were 43.8%, 45 days after treatment, with a baseline prevalence of 84% [31], 34.4%, three to four months after treatment, with a baseline prevalence of 44.9% [32], and 8.6%, eight weeks after treatment, with a baseline prevalence of 47.4% [33]. Therefore, the 36% prevalence in the second year of the study, which was conducted six months after MDA with praziquantel, signifies that the re-infection rate in this area remained relatively high although cure rate after MDA was not examined.

In terms of knowledge about schistosomiasis, many pupils were able to choose correct answers for signs, transmission mode, and prevention methods. However, even though *S. hae-matobium* is rarely found in the study area [26], more than half of pupils chose "blood in the urine" as a sign of dominant schistosomiasis in Mbita. In Kenya, *S. haematobium* and *S. mansoni* exist, and both species can be found along the shores of Lake Victoria [4] except in Mbita area. Residents along the shore of Lake Victoria call schistosomiasis "*layo remo*" which means "urinating blood" in local language; therefore, there might be a misconception among the



https://doi.org/10.1371/journal.pntd.0007572.g002

residents. However, it is unknown how popular the term is among children. Moreover, one of the science textbooks which was being used in this area only lists the sign of urinary schistosomiasis. This may also be one of the reasons why pupils chose "blood in urine" as a main sign of dominant schistosomiasis in Mbita. Previous studies in Tanzania [18] and Swaziland [17] also reported that signs of intestinal schistosomiasis were less known than urinary schistosomiasis among pupils, thus there might be a tendency that signs of *S. haematobium* are more known than that of *S. mansoni*.

In this study, the participants showed very good knowledge on schistosomiasis. This may be influenced by the Kenyan social and educational context. In Kenya, health is not a subject included in the primary education curriculum, but the Kenyan Ministry of Education encourages the inclusion of health topics in general subjects such as science, social studies, and language classes. The pupils learn about schistosomiasis in science class. Moreover, all pupils in Kenya who are supposed to graduate from primary schools must take a national examination, called the Kenya Certificate of Primary Education (KCPE), which affects their future education opportunity. Thus, they study hard, so it might be a reason why they have good knowledge of schistosomiasis because it was taught in classes. A study in Uganda [34] reported that health education using a child-friendly comic booklet failed to improve children's knowledge. However, our study showed that even though special health education activities were not conducted, pupils can get useful information from general classes when both the method and the contents are appropriate to the context of each country or area. Moreover, the participants



Fig 3. Proportion of schools using water fetched by pupils and fetching points (N = 93).

https://doi.org/10.1371/journal.pntd.0007572.g003

mentioned that the main source of information on schistosomiasis was teachers. It indicates that schools have a huge potential to distribute health information to the most susceptible age group as some previous studies reported [14,18]. Previous studies among adults have also reported that health workers, family, friends or neighbours are their source of information [15,16]. In the said studies, it is unknown which family member is the information source; however, other studies indicated that school children could pass health information to family and community members [35-37]. Therefore, appropriate health education in schools becomes very important and a useful tool for disease control.

Conflicting results in terms of attitude was found in the study. Nearly 75% of pupils thought that Lake Victoria is a source of schistosomiasis and half of pupils thought that they are not at risk of getting schistosomiasis. Yet, most of them come into contact with the lake water almost every day, as reflected by their answers to the questions on practices. Additionally, analyses of *Schistosoma* infection and practices at Lake Victoria showed a statistically positive association in accordance with previous studies conducted along the shores of Lake Victoria [26,38,39]. Although health education may succeed in giving correct information about schistosomiasis transmission, the demands of daily life and lack of safe alternatives seem to necessitate contact with the schistosome-infested water. Positive association between bathing in the lake and attitude on infection opportunity in daily life was also found, implying that pupils who bathe more frequently in the lake thought that they are more exposed to schistosomiasis than those who bathe less frequently in the lake. It also indicated that even though half of them did not agree that there is a possibility of them having a *Schistosoma* infection in daily life, they are aware that Lake Victoria is a source of infection and they go to the lake almost every day.

School children in the study area are taught in science class that one of the ways to prevent schistosomiasis is to avoid contact with contaminated water; however, most schools asked

pupils to fetch/bring water from the lake for school life, which exposes them to infection. This contradiction occurred because there is no safe water source for schools and this challenge must be overcome for successful behaviour change. Some teachers said that they do not have any alternative to secure water, so they advised pupils to shorten the contact time with water as much as possible. Since free education is a government policy in Kenya, it is very difficult for schools to ask parents to contribute to provision of safe water. The school health project conducted by Nagasaki University and funded by JICA installed rainwater-harvesting tanks in all public schools that did not have the tank; however, water can be secured through this method only during the rainy season. Thus, efforts should be made to appeal to the government to install piped water or boreholes in schools. Moreover, a new innovative method to secure water is needed. Another challenge is toilet use. The rate of toilet use at home is clearly lower in comparison to toilet use in schools because some households do not have a toilet, thus children could not avoid open defecation. Moreover, open defecation is still common in this area and is not a disgraceful behaviour. Similar to a report from a different area in Kenya, this is a social and psychological challenge that must be overcome for behaviour change [23].

The current main preventive measure for schistosomiasis in Kenya is school-based mass drug administration; however, despite the government's efforts, it has been intermittent due to logistical and financial challenges. In fact, school-based deworming was conducted four times in the study area during the past five years–in October 2012, June 2013, May 2015, and February 2016. However, praziquantel was not administered in May 2015, and although praziquantel was administered in the other years, its use was selective and very limited due to drug shortage. For example, during the MDA in February 2016, 88 (39%) out of 224 schools which conducted MDA used praziquantel in Mbita Sub-county. The keys for successful preventive chemotherapy (PCT) are continuous and more frequent drug administration [40,41]. There is a report that high prevalence of schistosomiasis was found after the withdrawal of school-based MDA [42].

The study also found that more than half of the pupils went to swim in the lake sometimes or every day. Thus, even if safe water will be provided, they might continue to go to the lake for recreation because of the hot weather in the area [43]. Moreover, fishing is the main source of income of residents in this area; hence fishermen are also at risk of schistosomiasis. The fishermen usually migrate from one fishing point to another for better catchment and this might make schistosomiasis control more complicated [44]. Since snails, which are the intermediate host, play an essential role in transmission [45], snail control is also important to prevent schistosomiasis [46].

This study has several limitations. First, urine samples were collected from each participant only once since *S. haematobium* is rare in the study area. Therefore, its prevalence might be underestimated. Second, the number of samples that were analysed for the prevalence of schistosomiasis became smaller than expected due to failure of some study participants to submit stool samples for three consecutive days. Third, since this study focused on school health, environmental factors such as distribution of snail species and *Schistosoma* infection among snails, which are important aspects for schistosomiasis control, were not documented.

This study confirmed that schistosomiasis is still highly prevalent among pupils in Mbita Sub-county, along the shore of Lake Victoria. Health education for pupils may be successful to acquire the correct knowledge; however, lack of safe water source and latrines blocked pupils' behaviour change. From the school health viewpoint, health education that reflects the social and cultural context of the community in the contents and teaching methods are needed in order to improve their knowledge and behaviour and to change the social norm. In addition to this, regular mass drug administration with monitoring, provision of safe water, and encouragement of toilet construction and use in both schools and community/households in cooperation with WASH activities are needed. Furthermore, an environmental survey to control snails is also needed. Finally, a comprehensive and innovative approach which harmonises central and local governments and other stakeholders, as well as community members is important to prevent schistosomiasis in this area. If the educational message is feasible, behaviour will be changed and the situation among children will be improved.

Supporting information

S1 Table. Contents of science textbook on schistosomiasis. (DOCX)

Acknowledgments

Many thanks to the officers of Ministry of Education, Science and Technology and Ministry of Health in Mbita and all our field staff for their dedicated work. We appreciate all the laboratory staff who conducted stool and urine tests and the Nagasaki University Mbita HDSS team for providing important data. We also thank the teachers and pupils who participated in this study. This work was conducted based on Kenya Research Station, Institute of Tropical Medicine, Nagasaki University (NUITM-KEMRI project). This paper is published with the permission of Kenya Medical Research Institute.

Author Contributions

Conceptualization: Jun Kobayashi.

Data curation: Rie Takeuchi.

Formal analysis: Rie Takeuchi.

Funding acquisition: Satoshi Kaneko.

Investigation: Rie Takeuchi.

Methodology: Rie Takeuchi.

Project administration: Rie Takeuchi, Sammy M. Njenga.

Supervision: Yoshio Ichinose, Satoshi Kaneko, Jun Kobayashi.

Visualization: Rie Takeuchi.

Writing – original draft: Rie Takeuchi.

Writing - review & editing: Rie Takeuchi, Sammy M. Njenga, Crystal A. Estrada.

References

- 1. World Health Organization. Working to overcome the global impact of neglected tropical diseases, First WHO report on neglected tropical diseases. Geneva, Switzerland: World Health Organization; 2010.
- Global Health Estimates 2014 Summary tables. World Health Organization, Geneva, Switzerland, 2014. Available from: http://www.who.int/entity/healthinfo/global_burden_disease/GHE_YLD_Global_ 2000_2012.xls.
- Colley DG, Bustinduy AL, Secor WE, King CH. Human schistosomiasis. Lancet. 2014; 383: 2253– 2264. https://doi.org/10.1016/S0140-6736(13)61949-2 PMID: 24698483
- Sang HC, Muchiri G, Ombok M, Odiere MR, Mwinzi PN. Schistosoma haematobium hotspots in south Nyanza, western Kenya: prevalence, distribution and co-endemicity with Schistosoma mansoni and soil-transmitted helminths. Parasit Vectors. 2014; 7: 125. https://doi.org/10.1186/1756-3305-7-125 PMID: 24667030

- World Health Organization. World Health Assembly Document. WHA3.26. Geneva, Switzerland: World Health Organization; 1950.
- 6. World Health Organization. World Health Assembly Document. WHA28.53. Geneva, Switzerland: World Health Organization; 1975.
- 7. World Health Organization. World Health Assembly Document. WHA29.58. Geneva, Switzerland: World Health Organization; 1976.
- 8. World Health Organization. World Health Assembly Document. WHA54.19. Geneva, Switzerland: World Health Organization; 2001.
- 9. World Health Organization. World Health Assembly Document. WHA65.21. Geneva, Switzerland: World Health Organization; 2012.
- 10. World Health Organization. The social context of schistosomiasis and its control. An introduction and annotated bibliography. Geneva, Switzerland: World Health Organization; 2008.
- 11. World Health Organization. Helminth control in school-age children: A guide for managers of control programmes. Geneva, Switzerland: World Health Organization; 2002.
- World Health Organization. Advocacy, communication and social mobilization for TB control: a guide to developing knowledge, attitude and practice surveys. Geneva, Switzerland: World Health Organization; 2008.
- **13.** World Health Organization. Bulletin of the World Health Organization. Geneva, Switzerland: World Health Organization; 2005.
- Midzi N, Mtapuri-Zinyowera S, Mapingure MP, Paul NH, Sangweme D, Hlerema G, et al. Knowledge attitudes and practices of grade three primary schoolchildren in relation to schistosomiasis, soil transmitted helminthiasis and malaria in Zimbabwe. Infect Dis. 2011; 11: 169.
- Dawaki S, Al-Mekhlafi HM, Ithoi I, Ibrahim J, Abdulsalam AM, Ahmed A, et. The Menace of Schistosomiasis in Nigeria: Knowledge, Attitude, and Practices Regarding Schistosomiasis among Rural Communities in Kano State. PLoS One. 2015 Nov 25; 10(11):e0143667. https://doi.org/10.1371/journal. pone.0143667 PMID: 26606264
- Mwai J, Njenga S, Barasa M. Knowledge, attitude and practices in relation to prevention and control of schistosomiasis infection in Mwea Kirinyaga county, Kenya. Public Health. 2016; 16(1): 819 <u>https://doi.org/10.1186/s12889-016-3494-y</u> PMID: 27538885
- Maseko TS, Mkhonta NR, Masuku SK, Dlamini SV, Fan CK. Schistosomiasis knowledge, attitude, practices, and associated factors among primary school children in the Siphofaneni area in the Lowveld of Swaziland. J Microbiol Immunol Infect. 2016 Jan 11. pii: S1684-1182(16)00004-9.
- Munisi DZ, Buza J, Mpolya EA, Angelo T, Kinung'hi SM. Knowledge, attitude, and practices on intestinal schistosomiasis among primary schoolchildren in the Lake Victoria basin, Rorya District, north-western Tanzania. Public Health. 2017; 17(1):731. https://doi.org/10.1186/s12889-017-4767-9 PMID: 28934944
- Mwandawiro CS, Nikolay B, Kihara JH, Ozier O, Mukoko DA, Mwanje MT, et al. Monitoring and evaluating the impact of national school-based deworming in Kenya: study design and baseline results. Parasit Vectors. 2013; 6:198. https://doi.org/10.1186/1756-3305-6-198 PMID: 23829767
- 20. Ministry of Education, Science and Technology and Ministry of Health, Kenya. Kenya National School-Based Deworming Programme. Year 1 (2012–2013) Results. Nairobi, Kenya: Ministry of Education, Science and Technology and Ministry of Health; 2014.
- USAID. Kenya, Water, sanitation and hygiene. Available from: https://www.usaid.gov/sites/default/files/documents/1860/Kenya_WASH_program_fact_sheet_2019.pdf Cited 19 May 2019.
- USAID. Water, sanitation and hygiene finance (WASH-FIN) Kenya. Available from https://www.usaid.gov/sites/default/files/documents/1860/WASH-FIN_fact_sheet_2019.pdf Cited 19 May 2019
- 23. UNICEF. End of programme evaluation–UNICEF Kenya WASH programme 2008–2014 End of programme evaluation report Final report. Available from https://www.unicef.org/evaldatabase/files/ UNICEF_Kenya_WASH-evaluation-report-final_with_TORs_Kenya_2016-014.pdf Cited 19 May 2019
- 24. World Vision Kenya. Annual report 2018. Available from https://www.wvi.org/sites/default/files/World% 20Vision%20Kenya%202018%20Annual%20Report.pdf Cited 19 May 2019
- Odiere MR, Rawago FO, Ombok M, Secor WE, Karanja DM, Mwinzi PN, et al. High prevalence of schistosomiasis in Mbita and its adjacent islands of Lake Victoria, western Kenya. Parasit Vectors. 2012; 5: 278. https://doi.org/10.1186/1756-3305-5-278 PMID: 23206383
- Nagi S, Chadeka EA, Sunahara T, Mutungi F, Justin YK, Kaneko S, et al. Risk factors and spatial distribution of *Schistosoma mansoni* infection among primary school children in Mbita District, Western Kenya. PLoS Negl Trop Dis. 2014 Jul 24; 8(7):e2991. doi: <u>10.1371/journal.pntd.0002991</u>. eCollection 2014 Jul. PMID: <u>25058653</u>

- Ng'etich AI, Rawago FO, Jura WG, Mwinzi PN, Won KY, Odiere MR. A cross-sectional study on schistosomiasis and soil-transmitted helminths in Mbita district, western Kenya using different copromicroscopic techniques. Parasit Vectors. 2016; 9: 87. https://doi.org/10.1186/s13071-016-1368-x PMID: 26883744
- 28. Kato T, Miura M. On the comparison of some stool examination methods. Jpn J Parasitol. 1954; 3: 35.
- Katz N, Chaves A, Pellegrino J. A simple device for quantitative stool thick-smear technique in Schistosomiasis mansoni. Rev Inst Med Trop Sa^o Paulo. 1972; 14: 397–400.
- **30.** World Health Organization. Prevention and control of schistosomiasis and soil-transmitted helminthiasis. Geneva, Switzerland: World Health Organization; 2002.
- Olds GR, King C, Hewlett J, Olveda R, Wu G, Ouma J, et al. Double-blind placebo-controlled study of concurrent administration of albendazole and praziquantel in schoolchildren with schistosomiasis and geohelminths. J Infect Dis. 1999; 179(4): 996–1003. https://doi.org/10.1086/314686 PMID: 10068597
- 32. Davis SM, Wiegand RE, Mulama F, Kareko EI, Harris R, Ochola E, et al. Morbidity associated with schistosomiasis before and after treatment in young children in Rusinga Island, western Kenya. Am J Trop Med Hyg. 2015; 92(5): 952–958. https://doi.org/10.4269/ajtmh.14-0346 PMID: 25758651
- 33. Kihara JH, Muhoho N, Njomo D, Mwobobia IK, Josyline K, Mitsui Y, et al. Drug efficacy of praziquantel and albendazole in school children in Mwea Division, Central Province, Kenya. Acta Trop. 2007; 102(3): 165–171. https://doi.org/10.1016/j.actatropica.2007.04.017 PMID: 17572368
- Stothard JR, Khamis AN, Khamis IS, Neo CHE, Wei I, Rollinson D. Health education and the control of urogenital schistosomiasis: Assessing the impact of the Juma Na Kichocho comic-strip medical booklet in Zanzibar. J Biosoc Sci. 2016; 48: S40–S55. <u>https://doi.org/10.1017/S0021932016000122</u> PMID: 27428065
- Onyango-Ouma W, Aagaard-Hansen J, Jensen BB. The potential of schoolchildren as health change agents in rural western Kenya. Soc Sci Med. 2005; 61(8): 1711–1722. <u>https://doi.org/10.1016/j.</u> socscimed.2005.03.041 PMID: 15967559
- Nonaka D, Kobayashi J, Jimba M, Vilaysouk B, Tsukamoto K, Kano S, et al. Malaria education from school to community in Oudomxay Province, Lao PDR. Parasitol Int. 2008; 57(1): 76–82. <u>https://doi.org/10.1016/j.parint.2007.09.005</u> PMID: 17980652
- Ayi I, Nonaka D, Adjovu JK, Hanafusa S, Jimba M, Bosompem KM, et al. School-based participatory health education for malaria control in Ghana: engaging children as health messengers. Malar J. 2010; 9: 98. https://doi.org/10.1186/1475-2875-9-98 PMID: 20398416
- Ruganuza DM, Mazigo HD, Waihenya R, Morona D, Mkoji GM. Schistosoma mansoni among preschool children in Musozi village, Ukerewe Island, North-Western-Tanzania: prevalence and associated risk factors. Parasit Vectors. 2015; 8: 377. <u>https://doi.org/10.1186/s13071-015-0997-9</u> PMID: 26178484
- Grimes JET, Tadesse G, Mekete K, Wuletaw Y, Gebretsadik A, French MD, et al. School Water, Sanitation, and Hygiene, Soil-Transmitted Helminths, and Schistosomes: National Mapping in Ethiopia. PLoS Negl Trop Dis. 2016 Mar 8; 10(3): e0004515. <u>https://doi.org/10.1371/journal.pntd.0004515</u> eCollection 2016 Mar. PMID: 26954688
- Ezeamama AE, He CL, Shen Y, Yin XP, Binder SC, Campbell CH Jr, et al. Gaining and sustaining schistosomiasis control: study protocol and baseline data prior to different treatment strategies in five African countries. BMC Infect Dis. 2016; 16:229. <u>https://doi.org/10.1186/s12879-016-1575-2</u> PMID: 27230666
- Ouedraogo H, Drabo F, Zongo D, Bagayan M, Bamba I, Pima T, et al. Schistosomiasis in school-age children in Burkina Faso after a decade of preventive chemotherapy. Bull World Health Organ. 2016; 94 (1): 37–45. https://doi.org/10.2471/BLT.15.161885 PMID: 26769995
- 42. Masaku J, Madigu N, Okoyo C, Njenga SM. Current status of Schistosoma mansoni and the factors associated with infection two years following mass drug administration programme among primary school children in Mwea irrigation scheme: A cross-sectional study. Public Health. 2015; 15: 739. https://doi.org/10.1186/s12889-015-1991-z PMID: 26231050
- Sanya RE, Tumwesige E. Elliott AM, Seeley J. Perceptions about interventions to control schistosomiasis among the Lake Victoria island communities of Koome, Uganda. PLoS Negl Trop Dis. 2017 Oct 2; 11(10):e0005982. https://doi.org/10.1371/journal.pntd.0005982 eCollection 2017 Oct. PMID: 28968470
- Standley CJ, Kabatereine NB, Lange CN, Lwambo NJS, Stothard JR. Molecular epidemiology and phylogeography of *Schistosoma mansoni* around Lake Victoria. Parasitology. 2010; 137: 1937–1949. https://doi.org/10.1017/S0031182010000788 PMID: 20561396
- 45. Standley CJ, Wade CM, Stothard JR. A fresh insight into transmission of schistosomiasis: A misleading tale of *Biomphalaria* in Lake Victoria. PLoS One. 2011; 6(10):e26563. https://doi.org/10.1371/journal. pone.0026563 Epub 2011 Oct 24 PMID: 22046308

46. Stothard JR, Campbell SJ, Osei-Atweneboana MY, Durant T, Stanton MC, Biritwum NK, et al. Towards interruption of schistosomiasis transmission in sub-Saharan Africa: developing an appropriate environmental surveillance framework to guide and to support 'end game' interventions. Infectious diseases and poverty. 2017; 6: 10. Infect Dis Poverty. 2017 Jan 14;6(1):10. <u>https://doi.org/10.1186/s40249-016-0215-9 PMID: 28088239</u>